Author’s response to reviews

Title: The global effect of maternal education on complete childhood vaccination: a systematic review and meta-analysis

Authors:

Jennifer Forshaw (jennifer.forshaw@doctors.org.uk)
Sarah Gerver (sgerver@gmail.com)
Moneet Gill (m1200394@sgul.ac.uk)
Emily Cooper (m1201289@sgul.ac.uk)
Logan Manikam (logan.manikam.10@ucl.ac.uk)
Helen Ward (h.ward@ic.ac.uk)

Version: 3 Date: 26 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for your comments following the resubmission of our manuscript, ‘The Global effect of maternal education on complete childhood vaccination: a systematic review and meta-analysis’. We are pleased to resubmit the updated manuscript with the following changes.

1. ‘Please ensure that all figures/tables and supplementary files, specifically Figure 2, are cited within the text. Any items which are not cited may be deleted by our production department upon publication.’

>>> Figure 2 has been referenced in line 183.

2. ‘Please provide figure titles/legends under a separate heading of ‘Figure Legends’ after the References. If Figure titles/legends are within the main text of the manuscript, please move them.’

>>> Figure legends have been taken out of the main manuscript and listed after the references at the end. Tables have been cited in the appropriate locations in the main body of text.

Thank you again for your feedback, please let us know if there are any further changes you would like us to make. We look forward to hearing your decision on our manuscript.
The authors declare that there are no conflicts of interest. All authors have approved the manuscript for submission. We confirm that the content of the manuscript has not been published, or submitted for publication elsewhere.

Kind regards,

The authors